Author's response to reviews

Title: Unusual MRI findings of dural arteriovenous fistula: Isolated perfusion lesions mimicking TIA

Authors:

Yong-Won Kim (yw.kim23@gmail.com)
Dong-Hun Kang (kdhdock@hotmail.com)
Yang-Ha Hwang (yangha.hwang@gmail.com)
Sung-Pa Park (sppark@knu.ac.kr)

Version: 2 Date: 16 February 2012

Author's response to reviews:

Dear editorial board of BMC Neurology,

With your request, the authors changed the paragraph for the consent section.

Consent

Written informed consent was obtained from the both patients for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal.

Sincerely yours,

Yang-Ha Hwang on behalf of the authors.